Author's response to reviews

Title: Protocol for Translabial 3D-Ultrasonography for diagnosing levator defects (TRUDIL): a multicentre cohort study for estimating the diagnostic accuracy of translabial 3D-ultrasonography of the pelvic floor as compared to MR imaging

Authors:

Kim JB Notten (kim.notten@mumc.nl)
Mirjam Weemhoff (m.weemhoff@mumc.nl)
Kirsten B Kluivers (k.kluivers@obgyn.umcn.nl)
Karlijn J Schweitzer (Schweitzer@alantvrouw.nl)
Femke Mulder (f.e.mulde@amc.uva.nl)
Jaap Stoker (j.stoker@amc.uva.nl)
Regina GH Beets-Tan (r.beets.tan@mumc.nl)
Jurgen J Futterer (j.futterer@rad.umcn.nl)
Roy FA Vliegen (RVN05@atriummc.nl)
Johannes LH Evers (jleh.evers@mumc.nl)
Gerold Link (gerold.link@mumc.nl)
Martin GM Bergmans (m.bergmans@lzd.nl)
Paul HNM Kampschoër (PKR01@atriummc.nl)
Ed TCM Gondrie (e.gondrie@orbisconcern.nl)
Iris van Gestel (ivgestel@viecuri.nl)
Ivo van Dooren (ima.van.dooren@sgweert.nl)
Carmen Dirksen (c.dirksen@mumc.nl)
Luc JM Smits (Luc.Smits@Epid.unimaas.nl)
Patrick M Bossuyt (p.m.bossuyt@amc.uva.nl)
Jan Paul WR Roovers (j.p.roovers@amc.uva.nl)

Version: 2 Date: 25 May 2011

Author's response to reviews:

Editor-in-Chief

BMC Women’s Health

Maastricht, 25 May 2011

Subject: Revisions "Protocol for Translabial 3D-Ultrasonography for diagnosing levator defects (TRUDIL): a multicentre cohort study for estimating the diagnostic accuracy of translabial 3D-ultrasonography of the pelvic floor as compared to MR imaging"

Dear Editor,

Again thank you for accepting our protocol for publication.
We have made several small revisions to our manuscript which we will discuss point-by-point.

Because this manuscript is a study protocol, the section discussion is not applicable. The abstract and manuscript where already submitted in the correct structure.

Concerning figure 1, we deleted this figure from the main text and we added the title of this figure after the reference list as you requested. We have also deleted the title and the figure number from this separately uploaded figure file.

Reference 3 is revised, a weblink was inserted.

We hope we have made all the essential changes necessary for publication.

Yours sincerely,

Kim NOtten, MD, PhD-student, corresponding author
Maastricht University Medical Center
Department of Obstetrics and Gynaecology
PO Box 5800
6200 WC Maastricht
The Netherlands
Tel: +31 433876762
email: kim.notten@mumc.nl